Author's response to reviews

Title: Application of biomedical informatics to chronic pediatric diseases: A systematic review

Authors:

Fatemeh Moeinedin (moeinedin@gmail.com)
Rahim Moineddin (rahim.moineddin@utoronto.ca)
Alejandro R Jadad (ajadad@ehealthinnovation.org)
Jemila S Hamid (jemila@utstat.toronto.edu)
Teresa To (teresa.to@sickkids.ca)
Joseph Beyene (joseph@utstat.toronto.edu)

Version: 4 Date: 23 April 2009

Author's response to reviews: see over
April 23, 2009

Formatting changes requested
-----------------------------
Major revisions (we require the author to make these changes).

Conclusions - Please insert a Conclusions section after the Discussion. This could contain the final paragraphs of the Discussion.

--> Done.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.

--> Done

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.

--> Done

Reference 32 - incorrect characters in last authors' name.

-→ Corrected.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. change capital A after the colon to lowercase a.

-→ Done.

Reference 21 - please use capital A at beginning of journal title (i.e. Analysis).

→ Done.

Table - please remove the text 'table 1 continued' from the top of the table pages, and remove the repeated column titles.

-→ Done.
Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

→ Done.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

→ Done.